Author's response to reviews

Title: Social inequalities in the participation and activity of children and adolescents with leukemia, brain tumors, and sarcomas (SUPATEEN): a protocol for a multicenter longitudinal prospective observational study

Authors:

Julia Roick (julia.roick@medizin.uni-halle.de)

Reinhard Berner (Reinhard.Berner@uniklinikum-dresden.de)

Toralf Bernig (toralf.bernig@uk-halle.de)

Bernhard Erdlenbruch (bernhard.erdlenbruch@klinikum-minden.de)

Gabriele Escherich (escherich@uke.de)

Jörg Faber (joerg.faber@uni-mainz.de)

Christoph Klein (Christoph.Klein@med.uni-muenchen.de)

Konrad Bochennek (konrad.bochennek@kgu.de)

Christian Kratz (kratz.christian@mh-hannover.de)

Joachim Kühr (Joachim.Kuehr@klinikum-karlsruhe.de)

Alfred Längler (Alfred.Laengler@uni-wh.de)

Holger Lode (lode@uni-greifswald.de)

Markus Metzler (markus.metzler@uk-erlangen.de)

Hermann Müller (mueller.hermann@klinikum-oldenburg.de)

Dirk Reinhardt (dirk.reinhardt@uk-essen.de)

Axel Sauerbrey (axel.sauerbrey@helios-gesundheit.de)

Florian Schepper (florian.schepper@medizin.uni-leipzig.de)

Wolfram Scheurlen (wolfram.scheurlen@nbg.diakonieneuendettelsau.de)

Dominik Schneider (Dominik.Schneider@klinikumdo.de)

Georg Schwabe (G.Schwabe@ctk.de)

Matthias Richter (m.richter@medizin.uni-halle.de)

Version: 1 Date: 21 Jan 2020

Author's response to reviews:

Ref. BPED-D-19-01521

Social inequalities in the participation and activity of children and adolescents with leukemia, brain tumors, and sarcomas (SUPATEEN): a protocol for a multicenter longitudinal prospective observational study

Dear Ms. Roick,

Your manuscript "Social inequalities in the participation and activity of children and adolescents with leukemia, brain tumors, and sarcomas (SUPATEEN): a protocol for a multicenter longitudinal prospective observational study" (BPED-D-19-01521) has been assessed. I am pleased to inform you that it is potentially acceptable for publication in BMC Pediatrics, once you have carried out some essential revisions:

** Thank you very much for considering our manuscript for publication in BMC Pediatrics.**

- -- Please ensure that all tables are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
- ** I added a citation to table 1 in the manuscript. All tables and figures are now cited.**
- --Please include a note in the funding statement that the funding body has peer-reviewed the ms as part of the award process.
- ** This note was added to the funding statement.**
- -- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
- ** The uploaded version does not contain any tracked changes, comments, highlight, strikethroughs or text in different colours. Tables and figures are also in clean versions.**
